Author's response to reviews

Title: Memory Aid - Computer Based Working Memory Training in Elderly with Mild Cognitive Impairment (MCI). A Randomized, Controlled Trial.

Authors:

  Marianne M Flak (marianne.moretro.flak@sshf.no)
  Susanne S Hernes (susanne.sorensen.hernes@sshf.no)
  Jon Skranes (jon.skranes@ntnu.no)
  Gro CC Lohaugen (gro.c.lohaugen@sshf.no)

Version: 5  Date: 4 April 2014

Author's response to reviews: see over
Dear Editors,

Thank you for your comments on the revised manuscript.

We have done the formatting requests:

- ensured that the title conforms to journal style for study protocol articles. The title now includes “study protocol for a randomized controlled trial.”. The text is altered and underlined in yellow, page 1.

- the date of the registration in Clinical trials.gov is added to the trial registration number at the end of the Abstract. Text altered and underlined in yellow, page 1.

Regards,

Marianne M. Flak
on behalf the authors